Author’s response to reviews

Title: Possible relation between consumption of different food groups and depression

Authors:

Gloria Grases (ggrases@cesag.org)
Maria Colom (macmesquida@gmail.com)
Pilar Sanchis (psanchis@ssib.es)
Felix Grases (fgrases@uib.es)

Version: 3 Date: 22 Feb 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for your e-mail. Following your instructions, I have made all the suggested changes:

1.- I provided a ‘Declarations’ heading after the Conclusion section.
2.- The ethical committee that approved the study is clearly specified in the ‘Ethical approval and consent to participate’ subsection.
3.- A supplementary files with the requested information has been included.
4.- In the Table, “p-value” has been changed to “p”, all the vertical lines has been removed and the horizontal lines has been also removed except the first two and the last.